Author's response to reviews

Title: Slum health: Diseases of neglected populations

Authors:

Lee W Riley (lwriley@berkeley.edu)
Albert I Ko (aik2001@med.cornell.edu)
Alon Unger (Alon.Unger@ucsf.edu)
Mitermayer G Reis (miter@cpqgm.fiocruz.br)

Version: 3 Date: 2 March 2007

Author’s response to reviews:

In this second revision, we added sentences at the suggestion of the reviewer to emphasize the point that the lack of data on health conditions in slums affects health planning and disease prevention by the formal health care system.